Author's response to reviews

Title: Solitary metastasis to the breast after complete resection of encapsulated type AB thymoma: a case report

Authors:

Pei-Wen Huang (hiruaa@yahoo.com.tw)  
Kuo-Ming Chang (a5523@ms7.mmh.org.tw)

Version: 2  
Date: 29 August 2014

Author's response to reviews:

To,

The Editor

Dear Sir,

We intend to publish an article entitled “Solitary metastasis to the breast after complete resection of encapsulated type AB thymoma: a case report” in your esteemed journal as a Case Report.

Type AB thymoma is generally considered as a benign tumor with excellent prognosis. Rare case reports of type A and AB thymomas with recurrence and metastasis have been documented, while the extrathoracic metastases are extremely rare. To our best knowledge, this is the first reported case of type AB thymoma with solitary metastasis to the breast 10 years after complete resection of non-invasive and encapsulated primary tumor.

Thanking you,

Yours' sincerely,

Pei-wen Huang

Corresponding contributor: Kuo-Ming Chang

a5523@ms7.mmh.org.tw